Author's response to reviews

Title: The Implications of Regional and National Demographic Projections for future GMS costs in Ireland through to 2026.

Authors:

Aisling Conway (aisling.conway@cit.ie)
Martin Kenneally (m.kenneally@ucc.ie)
Noel Woods (n.woods@ucc.ie)
Andreas Thuemmel (andreas.thuemmel@h-da.de)
Marie Ryan (marie.ryan@ucc.ie)

Version: 6 Date: 1 September 2014

Author's response to reviews:

Dear Editor,

I have now included an explanation why the study did not require ethical approval. It is included under the Methodology section at the end of the first paragraph. I hope this meets with your satisfaction and I look forward to any future correspondence.

Kind Regards,

Aisling